Author's response to reviews

Title: Transgenically-expressed secretoglobin 3A2 accelerates resolution of bleomycin-induced pulmonary fibrosis in mice

Authors:

Yan Cai (yan.cai@nih.gov)
Mitsuhiro Yoneda (yonedam@mail.nih.gov)
Takeshi Tomita (tomitat@research.twmu.ac.jp)
Reiko Kurotani (kurotanir@yz.yamagata-u.ac.jp)
Minoru Okamoto (mokamoto@rakuno.ac.jp)
Taketomo Kido (kido@iam.u-tokyo.ac.jp)
Hiroyuki Abe (abeh@yz.yamagata-u.ac.jp)
Wayne Mitzner (wmitzner@jhu.edu)
Arjun Guha (aguha@bu.edu)
Shioko Kimura (kimuras@mail.nih.gov)

Version: 3
Date: 29 May 2015

Author's response to reviews: see over
Dear Editor:

We wish to submit our re-revised manuscript entitled “Transgenically-expressed secretogoblin 3A2 accelerates resolution of bleomycin-induced pulmonary fibrosis in mice” for publication in the BMC Pulmonary Medicine.

First of all, we are very pleased to learn that our manuscript can be acceptable pending on our final minor revisions.

The reviewer 2 would like to have the detailed info on co-immunofluorescence images shown in Figure 1G and Supplemental Figure S2E. Accordingly we modified the figures, which now show the antibodies and the colors used.

I sincerely hope that our manuscript can now proceed to the next process for publication in the BMC Pulmonary Medicine.

Sincerely,
Shioko Kimura